Dear Reviewers,

Thank you for your time reviewing our manuscript. We appreciate your time and have made revisions. We have added a table summarizing the important points from the randomized controlled trial included in the review, as well as clarified some points in our discussion. Importantly, we note that there is no unified computer aided detection system in use, but several different systems. While the use of CADe certainly shows improved adenoma detection, improved accuracy and improved quality of colonoscopy, at this point it lacks external validation. As suggested, we have included the paper by Su et al. This paper showed that not only did their CADe system increase adenoma detection rate, but their group built a system that improves the quality of colonoscopy by alerting endoscopists to slow their withdrawal when appropriate. This suggests that there are even more potential advances to be made in CADe with regards to colonoscopy, however to validate this technology more widespread implementation and studies are required.

Thank you again for your time and attention to our manuscript.